



Xiaofang Jin 🔍, Azizah Omar * 🗅 and Kai Fu

School of Management, Universiti Sains Malaysia, Gelugor 11700, Malaysia; bbq6ay@gmail.com (X.J.); usmkai628@gmail.com (K.F.)

* Correspondence: aziemar@usm.my

Abstract: This study aimed to develop a conceptual framework that integrates both consumer-specific and product-specific factors, with a particular focus on consumer value perceptions regarding recycled apparel and their impact on consumer attitudes and purchase intention. The framework combines the Perceived Consumer Value (PCV) with the Knowledge–Attitude–Behavior (KAB) model. To validate the proposed conceptual framework, an online survey was conducted with a purposive sample of 716 participants in China. Using structural equation modeling, the hypotheses of the conceptual model were tested. This study identified four dimensions of perceived value and environmental knowledge as significant factors that positively influenced customers' attitudes toward purchasing recycled apparel. Notably, it was found that perceived quality value had a more pronounced effect on purchase intention than other factors. Interestingly, product knowledge did not demonstrate a significant impact on attitudes. By exploring this under-researched topic, this study provides new insights into consumer behavior in the context of sustainable fashion in a non-Western context.

Keywords: recycled apparel; environmental knowledge; product knowledge; perceived value; attitude; purchase intention

1. Introduction

In recent years, the emphasis on sustainable development and the need for eco-friendly consumption habits have grown markedly across societal and academic domains. This increased attention has highlighted the role of the fashion sector, particularly the fast fashion industry. "Fast fashion" is characterized by a shortened fashion cycle, which is associated with cheap manufacturing, more frequent consumption, and the short-term use and disposal of apparel items [1–3]. This trend has promoted the formation of rapid response systems and flexible supply chains in the apparel industry. However, it has also resulted in a considerable waste of resources and serious environmental pollution problems in the fashion industry [1–3].

Ikram [4] highlighted the negative social and environmental effects of the fashion industry, which was supported by Centobelli et al. [5], who noted that the fashion industry accounts for a staggering 10% of worldwide carbon emissions, making it a major pollution source. It is also responsible for 20% of the global industrial wastewater, at approximately 79 trillion liters annually, and contributes substantially to marine microplastic pollution. With over 92 million tons of textile waste produced annually, as pointed out by Musova et al. [1], the industry's detrimental impact is undeniable.

As the largest developing economy, China plays a prominent role in the global apparel industry, and its apparel market has been projected to reach a value of USD 615 billion by 2025 [6]. The textile sector has been instrumental in economic growth and public welfare but at a significant environmental cost. As per the China National Textile and Apparel Council (2021) [6], it is the third-largest source of wastewater effluents, contributing 17% in 2020, and accounts for 10% of all industrial carbon emissions. More importantly, China is



Citation: Jin, X.; Omar, A.; Fu, K. Factors Influencing Purchase Intention toward Recycled Apparel: Evidence from China. *Sustainability* 2024, *16*, 3633. https://doi.org/ 10.3390/su16093633

Academic Editors: Muhammad Mohiuddin, Slimane Ed-Dafali, Saeb Farhan Al Ganideh, Bilal Khalid and Ting Chi

Received: 3 March 2024 Revised: 12 April 2024 Accepted: 23 April 2024 Published: 26 April 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). also concerned about the environmental impact of textile waste. Currently, approximately 26 million tons of textile waste are generated in China every year, and it is predicted that this will increase in the future.

To counteract the pattern of unsustainable practices in fashion, there has been an emphasis on adopting the principles of sustainable development. According to the 14th Five-Year Plan for Economic and Social Development of the People's Republic of China (2021–2025), ambitious targets have been set for the textile industry, including reductions in energy consumption, carbon dioxide emissions, and water usage, alongside the goal of recycled fiber use [6].

As a sustainable solution, recycled clothing continues to attract the attention of scholars, the fashion industry, and environmentally conscious individuals. This trend is motivated by growing environmental concerns about resource availability and the increase in waste volume [7]. Recycled clothing, as highlighted by Wagner and Heinzel [8], is produced by breaking down textile waste into raw materials, which are then used in the production of new clothing items. It has been increasingly recognized as a promising and sustainable method for reducing material and energy use and ensuring sustainable production and consumption in the textile and apparel industries [7,8]. Due to these benefits, great interest in recycled fashion has emerged within academia and the fashion industry in recent years [6,9].

Amid the growing environmental awareness, Chinese textile and apparel companies are increasingly committing to environmental protection as part of their corporate social responsibilities. These efforts are largely focused on recycling waste clothing, with significant investments from fashion companies in recycling technologies. Notable collaborations, including those of LINING, ANTA, and LETIN, have led to advances in converting old clothes into high-quality recycled polyester fibers for new products [6].

Despite these measures, the uptake of recycled apparel among Chinese consumers is limited, underscoring a deficiency in market acceptance. While previous studies investigating recycled fashion in China have largely focused on design analysis and technological improvements in the fashion industry, empirical research on consumer purchasing behaviors regarding recycled fashion is limited, as noted by Jung et al. and Su et al. [10,11]. Therefore, it remains unclear which factors can encourage Chinese consumers to choose recycled clothing.

In previous studies on recycled fashion worldwide, the focus has been on the influencing factors, such as environmental concerns, consumers' pro-environmental attitudes, perceived quality, perceived value, and perceived risk [12–14]. However, it is unclear which specific aspects are considered by consumers when assessing the value of recycled apparel, especially in China, where the recycled apparel business is still at an early stage. Consumers lack knowledge about recycled apparel [6], and there are knowledge gaps regarding the fashion industry's environmental impacts, as highlighted by Jung et al., Lu et al., and Kim et al. [10,14,15], which may influence their purchase intention toward recycled clothing.

Therefore, this study seeks to explore the factors that cause Chinese consumers to choose recycled clothing, aiming to provide a corresponding framework and policy suggestions for the promotion of recycled fashion consumption in China. The main research questions are as follows.

RQ1: How does environmental knowledge influence Chinese consumers' attitudes toward recycled clothing? This question explores the impact of environmental awareness on consumer choices regarding sustainable fashion.

RQ2: What role does product knowledge play in shaping the attitudes of Chinese consumers toward recycled clothing? This question examines how familiarity with the attributes of recycled clothing can affect consumer preferences.

RQ3: How do different types of perceived value influence the attitudes and purchase intentions of Chinese consumers toward recycled clothing? This question aims to understand how consumers weigh the benefits (emotional, economic, quality, and social) of

recycled clothing against the potential drawbacks, influencing their decisions to purchase such a clothing item.

By identifying the factors associated with the acceptance of recycled apparel, this study proposes strategies to encourage sustainable consumption, thus aiding in reducing the fashion industry's environmental impacts and fostering sustainable market evolution. This research, therefore, aims to bolster the adoption of recycled apparel in China, contributing to a more sustainable fashion industry by addressing both theoretical and practical challenges, offering insights into consumer behavior, and suggesting strategies to increase the acceptance and utilization of recycled apparel.

The remainder of the paper is structured as follows: Section 2 provides a comprehensive literature review and a description of the development of hypotheses; Section 3 introduces the research methodology; Section 4 presents the results of the empirical investigation in detail; and Section 5 provides a discussion of all findings. Section 6 puts forward the study's theoretical contributions and managerial implications, followed by the study limitations and directions for future research in Section 7.

2. Literature Review and Hypothesis Development

2.1. Research on Recycled Apparel Consumption

With the rise of recycled apparel, scholars have shown increasing interest in its acceptance among consumers. Botwinick and Lu [12] found that clothing produced using recycled materials was often considered inferior in quality and less durable and functional than regular new clothing. Likewise, although consumers regard clothing made from recycled materials as more eco-friendly, no clear evidence has shown that they are willing to pay for such products given the products' clear functional or aesthetic shortcomings. From a Turkish perspective, Sener et al. [13] discovered that the perceived benefits derived from emotional, social, and epistemic values are insufficient in positively shaping consumers' attitudes toward recycled clothing. However, conditional value, rooted in factors such as environmental degradation and price incentives, exerts a significant positive influence on consumers' attitudes. Exploring the South Korean context, Kim et al. [14] determined that aesthetic and sanitary risks are notable deterrents in promoting positive product attitudes and purchase intentions toward recycled clothing. According to Meng and Leary [16] and Pretner et al. [17], consumers could also be wary of the potential health risks of consuming clothing made from recycled materials, given the unknown sources of the textile waste used and the recycling process. Within the Indian context, Chaturvedi et al. [18] identified willingness to pay, environmental concern, perceived value, and personal norms as critical drivers of Generation Z's purchasing behaviors regarding recycled clothing. Additionally, Park and Lin [19] found that even consumers with a positive attitude toward sustainable products often purchase clothing based on conventional factors, such as price, perceived value, and other functional benefits.

In summary, studies consistently show that consumers commonly see clothing made from recycled materials as relatively inferior in quality and value, with product safety concerns. In addition, consumers are not willing to sacrifice other benefits, such as quality, simply for the sake of sustainability. Finally, there is a lack of research concerning the purchase intention of Chinese consumers toward recycled fashion.

Thus, the sole reliance on highlighting the environmental contributions of recycled apparel is not sufficient to attract consumers. Addressing consumers' concerns about the value of recycled clothing is crucial. Firstly, it is necessary to identify the specific aspects considered by consumers when assessing the value of recycled apparel as being inadequate. Understanding these particular concerns is key in enabling sellers to provide tailored information, which could significantly reduce or eliminate these concerns. At the same time, consumer-specific factors, such as consumers' environmental knowledge and product knowledge, should be addressed, as consumers' decisions are not only influenced by their evaluation of the value of recycled clothing. Additionally, conducting research specifically addressing the purchasing behaviors and attitudes of Chinese consumers toward recycled apparel is imperative for a more logical and informed approach in this sector.

2.2. Theoretical Background

Kallgren and Wood's Knowledge–Attitude–Behavior (KAB) model [20], initially developed in 1986 to elucidate the influence of knowledge on behavior and its role in driving gradual change, has been widely applied in diverse academic fields, such as climate-smart food products [21], the circular economy [22], green product purchasing [23], education [24], and clinical services [25]. This model asserts that consumers' pursuit of knowledge culminates in either a positive or negative attitude toward a particular behavior. Central to the KAB model is the idea that attitudes are shaped by external factors and consumer knowledge, allowing for the seamless integration of consumer concerns related to external factors into the model. However, the KAB model has faced criticism for its simplicity. Research indicates that sustainable consumption is not only influenced by consumer-specific factors, such as environmental knowledge and product knowledge, but other factors associated with product perception also influence consumer behavior.

Thus, in addition to focusing on consumer-specific factors, a considerable body of research has also explored the impact of product-specific factors on purchase intention. These studies highlight the relevance of product-related concepts such as utility and value in explaining consumer behavior. Specifically, the Perceived Consumer Value (PCV) postulated by Sheth et al. [26], which underscores the critical role of perceived value in shaping behavioral intentions, has been influential in this regard. According to the research by Pereira et al. and Sebastián-Morillas et al. [27,28], perceived value is pivotal in determining consumer behavior. From this, we can see that perceived value is the subjective feeling of customers, involving the balance of interests and the emotional experience of customers. However, the perceptions of products or services vary based on a combination of needs, experiences, attitudes, and individual traits, suggesting that perceived value is a multifaceted construct encompassing various aspects, and it is applicable to areas such as sustainable fashion products [29], food delivery platform services [30], and live-streaming e-commerce [31]. Earlier, Sheth et al. [26] had divided the dimensionality of perceived value into five dimensions (functional value, conditional value, social value, emotional value, and cognitive value). Sweeney and Soutar [32] improved it and developed the classic four-dimensional scale (quality value, emotional value, price value, and social value). We employed Sweeney and Soutar's classic classification method for this research.

Our approach to exploring consumer behavior toward recycled apparel leverages the combined strengths of the KAB model and the PCV. These frameworks are selected for their proven effectiveness in discerning consumer behavior, especially in the realm of sustainable products. The KAB model, supported by research from scholars such as Polonsky et al. and Taufique et al. [33,34], addresses the critical gap between knowledge and subsequent action, which is particularly pertinent in the context of recycled apparel, where consumer decisions are heavily influenced by environmental awareness and attitudes.

Complementing the KAB model, the PCV delves into diverse types of consumption value, such as quality value, emotional value, price value, and social value that can guide consumer choices in the sustainable fashion sector, as noted by Busalim et al., Chi, and Kim et al. [14,35,36]. This layered approach is crucial in decoding the complexities of consumer decision-making processes in the sustainable or circular fashion context. The integration of these models serves to bridge the well-documented attitude–behavior gap that is prevalent in sustainable consumption practices, as indicated by Diddi et al. and Okur et al. [37,38]. By tracing the pathway from environmental knowledge to attitudes and then to behaviors while also considering the impact of various consumption values on these attitudes and behaviors, our research offers a comprehensive perspective on the dynamics at play.

Distinct from theories such as the Theory of Planned Behavior and the Theory of Reasoned Action, the fusion of the KAB model and the PCV uniquely addresses both the cognitive aspects (knowledge and attitudes) and affective components (values and

emotional connections) of consumer behavior in sustainable fashion, as mentioned by Riskos et al. and Tanrikulu [23,39]. This nuanced approach is especially relevant given the unique challenges within the recycled apparel market and is validated by its practical applicability in addressing contemporary market challenges, notably in studies focusing on the Chinese market [35,40], where sustainable fashion consumption is hindered by limited consumer knowledge and perceptions of value.

By combining the KAB model with the PCV, we aimed to provide an in-depth analysis that surpasses the insights provided by either framework independently, enabling a holistic understanding of how knowledge and values collectively shape consumer behavior toward recycled apparel. This strategy effectively addresses both the cognitive and affective dimensions of consumer behavior, offering a detailed and comprehensive exploration of consumer tendencies within the sustainable fashion domain. This integration not only addresses existing research gaps but also aptly captures the complexities of sustainable fashion consumption, highlighting our model's capacity to thoroughly examine the factors influencing consumer behavior in the recycled apparel sector.

2.3. Attitudes (ATT) and Purchase Intention (PI)

Purchase intention is a key focus in marketing research, linking attitudes with behavior to forecast consumer actions, as indicated by Sun et al. [41]. It was defined by Peña-García et al. [42] as the likelihood of a consumer buying a product after a thoughtful evaluation. Ajzen [43] defined intention as an individual's willingness to exert effort toward enacting a behavior. Researchers such as Onofrei et al. and Alwan et al. [44,45] have emphasized the importance of purchase intention in predicting future consumer choices, particularly in the realm of sustainable consumption. However, the variability of purchase intention leads to challenges in ensuring the robustness and consistency of findings related to attitudes and the prediction of purchase intention, as highlighted by the research of Maria et al. and Ajzen and Fishbein [46,47].

In addition, attitude plays a pivotal role in predicting intention, as posited by Ajzen [48], who describes it as a psychological tendency demonstrating favor or disfavor toward a particular behavior. This concept is essential in behavioral theories such as the Theory of Reasoned Action (TRA) and the Theory of Planned Behavior (TPB), which aim to understand human actions [49,50]. However, the effectiveness and universal applicability of these theories, particularly in a diverse range of consumer scenarios and across varying cultural environments, have generated considerable debate and examination. This leads to pressing questions regarding the global applicability and adaptability of these models across different contexts and cultural frameworks.

Despite the recognized impact of consumer attitudes on sustainable fashion product purchasing decisions, the existing literature has largely focused on Western cultural settings, such as those in America and Europe, leaving a significant gap in knowledge regarding other cultural contexts, including the Chinese cultural context, as documented by Mandarić et al., Wang et al., and Jain and Mishra [51–53]. Furthermore, research specifically addressing the purchase of recycled clothing remains notably scarce.

By addressing these knowledge gaps and the need for cultural diversity in research, the current study aims to explore the complex relationship between attitudes and purchase intentions within the context of recycled clothing consumption in China. This study proposes the following hypothesis:

H1: *Attitudes toward purchasing recycled apparel significantly affect purchase intention.*

2.4. Environmental Knowledge (EK) and Attitudes (ATT)

Research indicates that a major obstacle to the adoption of sustainable fashion, especially recycled apparel, is a lack of awareness and knowledge. Individuals do not fully understand how their fashion choices impact the environment, as outlined by Moon et al. and Muposhi and Chuchu [54,55], with many failing to see the negative effects of fast fashion, as indicated by Fang [56]. This ignorance contributes to a lack of concern about wearing unsustainable clothing, according to Breen [57]. Thus, it is vital to understand how knowledge impacts the purchasing of recycled apparel, as highlighted by Abbate et al., Abrar et al., and Hassan et al. [58–60].

Previous studies have explored how environmental knowledge affects eco-friendly actions. However, as reported by Hossain et al. and Pratiwi et al. [61,62], while knowledge about environmental damage can increase individuals' likelihood of displaying eco-friendly behaviors, this does not always happen. Gkargkavouzi et al. [63] found that knowledge has weak or no effects on the purchase of sustainable products. These inconclusive and fragmented results regarding the association between environmental knowledge and sustainable consumption mean that continued exploration is required by scholars and researchers, as mentioned by Amoah and Addoah [64].

In our study, we focus on environmental knowledge specific to recycled apparel, seeking to understand consumers' knowledge of how recycled apparel affects the environment; this knowledge has been linked to positive attitudes toward purchasing recycled apparel. Considering these insights, this study proposes the following hypothesis:

H2: Environmental knowledge significantly influences attitudes toward purchasing recycled apparel.

2.5. Product Knowledge (PK) and Attitudes (ATT)

Product knowledge represents the extent of a consumer's understanding regarding the various attributes of a specific product, which is shaped not only by their direct experiences but also by targeted marketing efforts [65–68]. It is an essential factor in the decision-making process, impacting perceptions of the product's inherent value and the associated risks of purchasing it [69]. However, most studies have primarily addressed aspects such as cost or green practices when examining product knowledge [70,71], leaving a gap in our understanding of how comprehensive product knowledge influences consumer attitudes, particularly toward specialized items.

Recent research efforts, such as those by Nurhayati and Hendar and Harahap et al. [72,73], have begun to shed light on the relationship between product knowledge and consumer attitudes. They underscore the importance of product knowledge in nurturing positive attitudes, which, in turn, significantly influence the adoption rates and purchase intention regarding items such as halal cosmetics and bubble tea drinks [74,75]. Despite these advances, current research remains narrowly focused on specific product categories, with insufficient attention given to how product knowledge affects attitudes toward environmentally friendly products, including recycled apparel.

This lack of a comprehensive understanding in the context of eco-conscious consumer goods underscores the need for further investigation. By acknowledging this gap, our study seeks to delve deeper into the field of sustainable fashion, specifically examining the impact of product knowledge on consumer attitudes toward purchasing recycled apparel. With the above considerations, this study posits the following hypothesis:

H3: Product knowledge significantly influences attitudes toward purchasing recycled apparel.

2.6. Perceived Value (PV), Attitudes (ATT), and Purchase Intention (PI)

Perceived value (PV) represents a consumer's comprehensive assessment of a product's benefits, juxtaposing the gains against the sacrifices made, as Park et al. and Zeithaml [19,76] have pointed out. While perceived value has traditionally been segmented into quality and price dimensions, this binary classification neglects nuanced aspects such as emotional and social factors, leading to an incomplete understanding of consumer value perceptions [77]. Notably, the model developed by Liu et al. [78] introduces a more complex framework, suggesting that consumer assessments include emotional, social, price, and quality factors. Our study's emphasis on the four dimensions of perceived value—emotional, social, quality, and price—is grounded in their well-documented influence on consumer purchasing decisions [79,80]. These dimensions encompass both the psychological aspects through emotional and social values and the tangible considerations of quality and price-weighted by consumers [81]. This distinction allows for an in-depth exploration of the myriad factors affecting consumer choices, enhancing the study's comprehensiveness.

Focusing on these specific dimensions enables a clear, detailed analysis by highlighting the unique influence of each on consumer behavior. This approach aligns with our aim to deconstruct the contribution of each dimension to perceived value, enriching our findings' clarity and depth. The selection of these dimensions is supported by a wealth of existing research, ensuring the relevance and applicability of our study's outcomes.

Opting to exclude dimensions such as environmental and epistemic values, which are critical to sustainable consumption, is a strategic choice intended to ensure that the research is focused on and tailored to our target consumer group and context, which might not emphasize these values to the same extent. However, we recognize their potential significance in other contexts or areas and encourage future studies to explore these dimensions for a fuller understanding of consumer decision-making, especially where these values are more critical.

Our research specifically addresses the Chinese market's demand for environmentally friendly apparel, prioritizing the dimensions that align with Chinese consumers' preferences, as indicated by Chi [36]. The analysis revealed a particular emphasis on social quality and price values, suggesting their cultural and market-specific importance in China's environmentally friendly apparel consumption. This focus reflects the unique cultural and market dynamics at play, justifying our research approach's targeted nature. This study, thus, aimed to discern the PV of recycled apparel through these multifaceted lenses.

Emotional resonance with products plays a pivotal role in shaping consumer preferences, as products are not merely functional items but also vehicles that elicit specific emotional responses [82,83]. Design elements are frequently crafted to elicit positive emotional reactions, contributing to the overall value that consumers derive from their purchases [84]. In the domain of sustainable fashion, aspects such as self-esteem and selfexpression take center stage, reflecting the emotional benefits that consumers expect from their eco-friendly purchases [85,86]. However, there is a notable dearth in the literature regarding how the emotional value derived from recycled products impacts consumer attitudes and their subsequent purchase intention. A previous study [87] on consumers' purchase of recycled products focused on the functional value but largely neglected the emotional value.

This oversight in previous research highlights a crucial area for further investigation, particularly within the sustainable fashion sector, which is increasingly being recognized for its potential to foster emotional connections through eco-conscious purchasing choices. The perceived emotional value (PEV) associated with recycled apparel, encompassing feelings of pride, satisfaction, or alignment with personal values, remains underexplored in terms of its effect on consumer behavior.

By addressing this gap, our study aimed to investigate the intricate relationship between the emotional value that consumers perceive in purchasing recycled apparel and how this influences their attitudes and decision-making processes toward these products. By reflecting on these insights, this study proposes the following hypotheses.

H4a: *Perceived emotional value (PEV) significantly affects attitudes toward purchasing recycled apparel.*

H4b: *Perceived emotional value (PEV) significantly influences purchase intention toward recycled apparel.*

Social value in consumer behavior encompasses the perceived benefits that a product provides in enhancing an individual's interactions within their social sphere, as indicated by Liu et al. [78]. Clothing, beyond its basic functionality, serves as a potent symbol of individuals' social status and identity, influencing and reflecting social boundaries and affiliations, as highlighted by Özdil [88]. This perspective underscores the importance of apparel as a means of social expression, allowing individuals to navigate and assert their positions within various social groups, a concept reinforced by Crane, Entwistle, and Barnard [89–91]. The role of clothing extends into the realm of social signaling, where it transcends mere utility to embody social values and norms.

The influence of social value on consumer decisions is particularly pronounced in the context of visible consumer goods, such as apparel and jewelry, as Sheth [26] has observed. This notion is supported by Chi [36], who found social value to be a significant factor in the perceived value of eco-friendly clothing. Similarly, the research by Gwozdz et al. [92] on sustainable fashion among young Swedes emphasized the social signaling value of clothing, noting its importance in the consumer choices of male participants, particularly in terms of projecting an image of success.

By drawing on these insights, it is reasonable to infer that consumers may also associate social benefits with the purchase of recycled apparel, aligning with contemporary environmental consciousness and societal values. Such purchases not only reflect personal environmental commitments but also enhance one's social image and acceptance, particularly in communities prioritizing sustainability. Based on these considerations, our study seeks to explore the impact of perceived social value on consumer attitudes and behaviors toward recycled apparel. Thus, we propose the following hypotheses:

H5a: *Perceived social value (PSV) significantly affects attitudes toward recycled apparel.*

H5b: *Perceived social value (PSV) significantly influences purchase intention toward recycled apparel.*

Sustainable products, particularly in the fashion sector, are frequently commended for their superior quality, embodying the sustainability ethos of durability and longevity [93]. Recycled apparel, a major component of the sustainable fashion industry, exemplifies this focus on quality, which has emerged as a fundamental element influencing consumer perceptions and driving eco-friendly consumption patterns [94]. This correlation between perceived quality and sustainable purchasing behaviors has been substantiated by findings from scholars such as Wasaya et al., Queiroz et al., Cheung et al., and Tang et al. [95–98].

The significance of quality and durability has been further underscored in the aftermath of the COVID-19 pandemic, which has seen shifts in consumer priorities toward more long-lasting products in response to economic uncertainties and increased financial prudence, as documented by Hasbullah et al. [99]. The concept of perceived quality value, reflecting the comparison between experienced quality and anticipated or expected quality [100], is crucial in shaping consumers' overall evaluations and judgments regarding a product's excellence.

In this context, recycled apparel that aligns with or surpasses consumers' expectations regarding quality can significantly influence their purchasing preferences, positioning these products as desirable alternatives within the market. This transition underscores the importance of perceived quality value as a key determinant of consumer preferences for recycled clothing. Given the current market dynamics and consumer trends, this study aimed to delve into the intricate relationship between perceived quality value and consumer attitudes toward recycled fashion. In view of these considerations, we propose the following hypotheses to examine this relationship further:

H6a: Perceived quality value (PQV) has a significant impact on attitudes toward purchasing recycled apparel.

H6b: *Perceived quality value (PQV) significantly influences purchase intention toward recycled apparel.*

Perceived price value refers to a consumer's evaluation of a product's cost relative to its perceived benefits. Studies by Jeaheng et al. and Singh and Alok [101,102] and extensive research conducted by [103–105] have consistently indicated that perceived price value significantly shapes consumer attitudes and intentions across various contexts. However, the willingness of consumers to pay a premium for green products remains a complex issue, particularly in developing nations. Notably, in countries such as China, reports suggest that consumers might not be as discouraged by higher prices if they believe in the functional value and quality of sustainable products, a sentiment reflected in the findings of the studies by Zhang and Ito and Zhang [81,106]. However, these findings do not entirely concur with those of Shen et al. and Bigliardi et al. [107,108], who observed that consumers tend to balance considerations of both price and quality when deciding whether to purchase recycled products.

By considering these findings, our study aimed to explore this dimension within the sustainable fashion industry, particularly focusing on recycled apparel. Therefore, we propose the following hypotheses to guide our investigation:

H7a: Perceived price value (PPV) has a significant impact on attitudes toward recycled apparel.

H7b: *Perceived price value (PPV) significantly influences purchase intention toward recycled apparel.*

model in Figure 1.

According to the hypotheses listed above, this study puts forward the theoretical



Figure 1. Proposed research model. EK for environmental knowledge, PK for product knowledge, PV for perceived value, PEV for perceived emotional value, PSV for perceived social value, PQV for perceived quality value, PPV for perceived price value, ATT for attitude, and PI for purchase intention.

3. Methodology

3.1. Questionnaire Design and Pre-Testing Study

This study utilized a survey questionnaire to collect data, beginning with demographic information before moving on to explore the study's key variables. We measured all latent variables using scales that have been previously validated. The respondents rated a series of statements on a five-point Likert scale, ranging from 1 (strongly disagree) to 5 (strongly agree), to provide detailed insights into their perceptions [109].

Prior to data collection, a pilot study ensured the questionnaire's validity and reliability. This phase involved a rigorous examination of the questionnaire's content, including the clarity of the items, the language used, the sequence, the format, and the complexity of the questions and instructions. To accurately reflect the participants' primary language, the questionnaire underwent a forward and backward translation process from English to Chinese [110].

With all measurement items achieving a loading factor above the 0.7 threshold, we decided to retain all 29 items for the in-depth analysis. The finalized survey items are detailed in Table 1.

Variable	Item Number	Item Content	Source	
	EK1	I know microfiber pollution is related to the textile and apparel industry		
Environmental Knowledge	EK2	I know the textile and apparel industry is a large contributor to greenhouse gas emissions	Kim and Damhorst (1998) [111]	
-	EK3	I know consumers contribute to the environmental issues associated with the apparel industry		
	PK1	I am quite familiar with recycled apparel		
Product	PK2	I know which brands sell recycled apparel	Kang et al. (2013) [112]	
Knowledge	PK3	I can easily determine where I can purchase recycled apparel		
	PEV1	Recycled apparel would make me feel good		
Perceived	PEV2	Recycled apparel would give me pleasure	M_{2} = h_{2} = h_{2} (2014) [112]	
Value	PEV3	Recycled apparel would make me want to use them	Waish et al. (2014) [115]	
	PEV4	Recycled apparel are the ones that I would enjoy		
	PSV1	Recycled apparel would improve the way I am perceived		
Perceived Social	PSV2	Recycled apparel would give their owner social approval	Walsh et al. (2014) [113]	
value	PSV3	Recycled apparel would make a good impression on other people		
	PSV4	Recycled apparel would help me to feel acceptable		
	PQV1	Recycled apparel is well-made		
During	PQV2	Recycled apparel has an acceptable standard of quality	-	
Quality	PQV3	Recycled apparel would last a long time	Walsh et al. (2014) [113]	
Value	PQV4	Recycled apparel has consistent quality		
	PQV5	Recycled apparel would perform consistently		
	PPV1	Recycled apparel would be economical		
Perceived	PPV2	Recycled apparel is reasonably priced	M_{2} = h_{2} = h_{2} (2014) [112]	
Value	PPV3 Recycled apparel offers value for money Walsh et al.		waish et al. (2014) [115]	
	PPV4	Recycled apparel is good for the price		

Table 1. The measurement scales and their sources.

Variable	Item Item Content		Source	
	ATT1	My attitude toward purchasing recycled apparel is very positive		
Attitude	ATT2	I think purchasing recycled apparel is very favorable for me	Preuit and Yan (2016) [114]	
	ATT3	I think purchasing recycled apparel is a pleasant choice		
	PI1	In the future, I intend to purchase recycled apparel		
Purchase Intention	PI2	In the future, I will tell my friends about the Preuit and Yan (20 recycled apparel		
	PI3	I will prioritize recycled over fast-fashion apparel		

Table 1. Cont.

3.2. Sampling and Data Collection

Data collection for our study was carried out between October 2023 and January 2024 utilizing the renowned online survey platform Wenjuanxing for survey distribution. Online surveys are notably efficient in terms of cost and time, and they enable a broad reach, making them an attractive option for social science researchers [115]. Wenjuanxing, also known as Questionnaire Star, is a particularly popular online survey tool in China that is widely used in market research, educational assessments, customer satisfaction surveys, and many other contexts. Having over ninety million users, it covers a broad spectrum of the population, including individuals of various ages, genders, professions, educational backgrounds, etc. From a statistical perspective, this extensive user base enhances the representativeness of participants recruited through Questionnaire Star.

We employed purposive sampling to select participants who were likely to provide valuable insights into the study's subject matter. We also adopted purposive sampling due to ethical considerations, which necessitated that the survey be distributed only to individuals above the age of 18. On this basis, the survey platform randomly sent the questionnaire to its registered members who were 18 years of age or older. We assured respondents that their responses would be treated confidentially and anonymously since the information was intended for publication in scientific journals.

Table 2 details the respondents' demographic information. The demographic data revealed a balanced gender distribution among the respondents, with males at 50.1% (359 respondents) and females at 49.9% (357 respondents). The age range of 26–35 years dominated (287 respondents, 40.1%), with decreasing representation in older and younger groups. Most respondents earned below 3000 units of currency (CNY) monthly, with the majority falling into the <1000 (28.8%) and 1000–3000 (32.2%) income brackets. Higher income levels were less prevalent.

Characteristic	Category	Ν	%
Gender	Male	359	50.1
	Female	357	49.9
Age	18–25	146	20.4
0	26-35	287	40.1
	36-45	195	27.2
	46-55	69	9.6
	56-65	16	2.2
	>65	3	0.4

Table 2. Demographic information of respondents.

Characteristic	Category	Ν	%
Income (CNY)	<1000	206	28.8
	1000-3000	231	32.2
	3001-5000	106	14.8
	5001-8000	81	11.3
	8001-10,000	41	5.7
	10,001–20,000	17	2.4
	>20,000	34	4.8
Educational Level	Junior high school	133	18.6
	Senior high school	189	26.4
	College diploma	235	32.8
	Bachelor's degree	140	19.6
	Master's degree	19	2.7
Occupation	Government employees	64	8.9
	Business managers	69	9.6
	Office workers	144	20.1
	Professionals	31	4.3
	Factory workers	90	12.6
	Salesperson	25	3.5
	Self-employed	111	15.5
	Agricultural workers	19	2.7
	Retired	23	3.2
	No occupation	28	3.9
	Current students	112	15.7

Table 2. Cont.

In terms of education, a substantial number of respondents held a college diploma (235 respondents, 32.8%) or a bachelor's degree (140 respondents, 19.6%), with fewer having completed only high school or attained a master's degree. In terms of occupation, office workers (144 respondents, 20.1%), self-employed individuals (111 respondents, 15.5%), and students (112 respondents, 15.7%) are the most common, with fewer respondents in various other professional and non-professional roles.

3.3. Statistical Analysis

For statistical analysis and structural equation modeling, we utilized the Smart PLS 4.0 software. The choice of partial least squares structural equation modeling (PLS-SEM) over covariance-based SEM (CB-SEM) was driven by several considerations:

- 1. The novelty and complexity of the research model: Our study explored a relatively new context where existing measurement models are not fully established or are inadequate. PLS-SEM is particularly advantageous in such new and complex scenarios, as it allows for the development and refinement of the measurement models in conjunction with the assessment of the structural model [116,117]. This dual capability was crucial for our research, which integrated the KAB model and the TCV.
- 2. The focus on theory development and prediction: Unlike CB-SEM, which is tailored toward theory confirmation and model fitting, PLS-SEM is better suited for theory development and exploratory research [118,119]. This distinction aligns with our study's objectives of theory expansion and predictive analysis, especially in the context of consumer behavior toward sustainable products.
- 3. Statistical power and model complexity: PLS-SEM provides robust statistical power even with smaller sample sizes and can handle complex models with many constructs and paths. This made it an ideal choice for our study, which involved multiple interrelated constructs within a comprehensive model [118]. Additionally, PLS-SEM does not impose the same stringent distributional assumptions as CB-SEM, which was beneficial given the nature of our data.

4. Prevalence in similar studies: PLS-SEM is used in many sustainable consumption studies. Works by scholars such as Yang(a) et al., Usman et al., Yang(b) et al., and Hair et al. [120–123] exemplify the application of PLS-SEM in sustainable consumption, reinforcing our methodological choice.

By providing these detailed justifications for our methodological approach, we hope to clarify our choice of PLS-SEM and demonstrate its alignment with the goals and data structure of our study.

4. Results

4.1. Measurement Model

This study rigorously evaluated the reflective measurement model, employing measures of reliability, convergent validity, and discriminant validity according to the guidelines by Hair et al. [124] and Memon et al. [125]. The confirmatory factor analysis results are summarized in Table 3, indicating factor loadings between 0.711 and 0.826, which surpassed the threshold of 0.60. The average variance extracted (AVE) for each construct exceeded 0.50, and the composite reliability ranged from 0.792 to 0.846, above the 0.70 threshold. The variance inflation factor (VIF) values ranged from 1.18 to 1.568, well below the maximum value of 3 suggested by Hair et al. [126], indicating the absence of collinearity issues and minimal common method bias. See Appendix A for a comprehensive series of detailed charts and graphical illustrations of the results.

Table 3. Reliability and validity of the study.

Construct	Item	Factor Loading	VIF	CR	AVE
-	EK1	0.733	1.18		
EK	EK2	0.708	1.231	0.792	0.56
	EK3	0.8	1.235	-	
	PK1	0.765	1.308		
РК	PK2	0.769	1.224	0.8	0.571
	PK3	0.732	1.201	-	
	PEV1	0.774	1.517		
DEV	PEV2	0.76	1.49	0.838	0 565
I L V	PEV3	0.713	1.301	- 0.000	0.505
	PEV4	0.757	1.459	-	
	PSV1	0.718	1.43		
PSV	PSV2	0.721	1.309	0 821	0 534
150	PSV3	0.744	1.398	- 0.021	0.004
	PSV4	0.738	1.402	-	
	PQV1	0.75	1.534		
	PQV2	0.711	1.47	-	
PQV	PQV3	0.72	1.426	0.846	0.524
	PQV4	0.715	1.44	-	
	PQV5	0.722	1.503	-	
PPV	PPV1	0.776	1.531	_	
	PPV2	0.717	1.422	0.836	0 561
	PPV3	0.726	1.371	- 0.000	0.001
	PPV4	0.774	1.568	-	

Construct	Item	Factor Loading	VIF	CR	AVE	
	ATT1	0.826	1.545			
ATT	ATT2	0.796	1.445	0.838	0.633	
	ATT3	0.763	1.281	-		
PI	PI1	0.786	1.378			
	PI2	0.761	1.235	0.824	0.609	
	PI3	0.795	1.417	-		

Table 3. Cont.

EK for environmental knowledge, PK for product knowledge, PEV for perceived emotional value, PSV for perceived social value, PQV for perceived quality value, PPV for perceived price value, ATT for attitude, and PI for purchase intention.

Henseler et al. [127] suggested that the heterotrait–monotrait (HTMT) ratio can further detect the lack of discriminant validity and is superior to the Fornell–Larker criterion, considering that it "offers the best balance between high detection and low arbitrary violation". Thus, this study used the HTMT ratio to evaluate discriminant validity; as reported in Table 4, all HTMT values were lower than the cut-off value of 0.85 [128].

Table 4. HTMT results.

ATT	EK	PEV	PI	РК	PPV	PQV	PSV
ATT							
EK	0.623						
PEV	0.827	0.696					
PI	0.832	0.713	0.838				
PK	0.801	0.757	0.712	0.846			
PPV	0.842	0.731	0.753	0.725	0.68		
PQV	0.744	0.716	0.821	0.799	0.818	0.808	
PSV	0.756	0.764	0.782	0.682	0.798	0.84	0.785

EK for environmental knowledge, PK for product knowledge, PEV for perceived emotional value, PSV for perceived social value, PQV for perceived quality value, PPV for perceived price value, ATT for attitude, and PI for purchase intention.

4.2. Structural Model

The validity of the structural model was confirmed using various metrics in this study. Key indicators, such as the coefficient of determination (R2) and predictive relevance (Q2), were evaluated by Hair et al. [124].

The model demonstrated moderate explanatory power, with an R2 value of 0.672 for PI and 0.648 for ATT, exceeding the 0.5 threshold set by Falk and Miller [129]. The predictive relevance was assessed using the Q2 values for exogenous variables, where PI (Q2 = 0.432) and ATT (Q2 = 0.297) both exhibited predictive relevance above 0, as per Geisser [130]. This confirmed the structural model's validity.

4.3. Hypothesis Testing

The hypotheses were tested using a bootstrapping method with 5000 subsamples, maintaining a 95% confidence level, as recommended by Memon et al., Purwanto and Witte and Witte [125,131,132].

When a path coefficient is non-zero, a t-statistic surpassing the critical value of 1.69 signifies statistical significance, as does a *p*-value falling below the commonly accepted threshold of 0.05. It points to a substantial relationship between the variables. The confidence interval provides a range suggesting where the true effect size likely falls, enhancing our understanding of the estimate's reliability and accuracy [132]. Moreover, a confidence interval excluding 0 (or 1) reaffirms this relationship's significance [133,134]. Together, these statistical metrics facilitate a thorough verification of the hypotheses, offering a detailed evaluation of the dynamics between the studied variables.

The detailed results of the hypothesis testing and their statistical significance are displayed in Figure 2 and Table 5. Attitudes toward purchasing recycled apparel (ATT) had a significant positive effect on purchase intention (PI) ($\beta = 0.273$, t = 5.427, *p* < 0.001), validating H1. Environmental knowledge (EK) showed a significant positive effect on attitudes toward purchasing recycled apparel ($\beta = 0.066$, t = 1.914, *p* < 0.05); therefore, H2 was supported. However, the relationship between product knowledge (PK) and attitudes toward purchasing recycled apparel (ATT) was not statistically significant, as the t-value was smaller than 1.69 and the *p*-value was larger than 0.05 ($\beta = 0.023$, t = 0.564, *p* > 0.05). Hence, H3 was not supported.



Figure 2. PLS-SEM output for path coefficients and *p*-values via bootstrapping procedure. EK for environmental knowledge, PK for product knowledge, PEV for perceived emotional value, PSV for perceived social value, PQV for perceived quality value, PPV for perceived price value, ATT for attitude, and PI for purchase intention.

Hypothesis Number	Path	Path Coefficient	T-Statistic	<i>p</i> -Value	LLCI (5.00%)	ULCI (95.00%)	Supported
H1	$\text{ATT} \rightarrow \text{PI}$	0.273	5.427	0	0.188	0.352	Yes
H2	$\text{EK} \to \text{ATT}$	0.066	1.914	0.028	0.01	0.124	Yes
H3	$\text{PK} \to \text{ATT}$	0.023	0.564	0.287	-0.039	0.092	No
H4a	$\text{PEV} \rightarrow \text{ATT}$	0.158	3.129	0.001	0.074	0.241	Yes
H4b	$\mathrm{PEV} \to \mathrm{PI}$	0.123	2.598	0.005	0.044	0.201	Yes
H5a	$\text{PSV} \to \text{ATT}$	0.13	2.239	0.013	0.031	0.226	Yes
H5b	$\text{PSV} \to \text{PI}$	0.134	2.665	0.004	0.053	0.219	Yes
H6a	$PQV \to ATT$	0.263	4.282	0	0.157	0.359	Yes
H6b	$\mathrm{PQV} \to \mathrm{PI}$	0.206	3.583	0	0.114	0.301	Yes
H7a	$\mathrm{PPV} \to \mathrm{ATT}$	0.267	4.706	0	0.175	0.364	Yes
H7b	$\mathrm{PPV} \to \mathrm{PI}$	0.176	3.293	0	0.087	0.264	Yes

Table 5. Hypothesis tests.

EK for environmental knowledge, PK for product knowledge, PEV for perceived emotional value, PSV for perceived social value, PQV for perceived quality value, PPV for perceived price value, ATT for attitude, and PI for purchase intention.

In H4, perceived emotional value (PEV) ($\beta = 0.158$, t = 3.129, p < 0.001), perceived social value (PSV) ($\beta = 0.13$, t = 2.239, p < 0.05), perceived quality value (PQV) ($\beta = 0.263$, t = 4.282, p < 0.001), and perceived price value (PPV) ($\beta = 0.267$, t = 4.706, p < 0.001) positively affected attitudes. Thus, H4a, H5a, H6a, and H7a were all supported.

In H5, perceived emotional value (PEV) ($\beta = 0.123$, t = 2.598, p < 0.01), perceived social value (PSV) ($\beta = 0.134$, t = 2.665, p < 0.05), perceived quality value (PQV) ($\beta = 0.206$, t = 3.583, p < 0.001), and perceived price value (PPV) ($\beta = 0.176$, t = 3.293, p < 0.001) positively affected purchase intention. Thus, H4b, H5b, H6b, and H7b were all supported. For detailed graphical representations of the SmartPLS results, please see Appendix A.

5. Discussion and Conclusions

The aim of our study was to investigate consumers' value and knowledge perceptions of recycled clothing. The results of the study offer several key insights.

First, this study underscores that consumers with positive attitudes toward recycled clothing are more likely to buy such items, seeing them as beneficial, useful, and responsible choices. This observation aligns with Jalil and Shaharuddin [135], who found a direct correlation between consumer attitudes and their willingness to purchase recycled clothing. This result confirms that consumers' attitudes are important in creating a positive purchase intention toward recycled fashion consumption, which is also suggested by theories such as the Theory of Reasoned Action (TRA), the Theory of Planned Behavior (TPB), and the KAB model. Indeed, this study confirms that in the emerging Chinese market for recycled clothing, consumers' attitudes continue to play a crucial role in shaping their purchase intention. It is essential for brands in this sector to understand and address the attitudes of their potential customers to foster a more sustainable approach to fashion.

Second, this study's findings suggest that environmental knowledge increases positive attitudes toward purchasing recycled clothing, which is consistent with the findings of previous research, such as the studies by Bong Ko and Jin [136] and Vicente-Molina et al. [137]. This result confirms that consumers' lack of knowledge can act as an obstacle to eliciting a positive attitude toward recycled clothing, with many consumers failing to translate their environmental awareness into action. As there have been contradictory findings in the past, such as those by Jaiswal and Kant [138] and Gkargkavouzi et al. [63], it was found that knowledge has weak or no effects on the purchase of sustainable products. Our finding thus adds clarity to the accumulated knowledge. It implies that consumers are aware that environmentally sustainable products help to conserve natural resources, which leads to the formation of an attitude that favors the use of environmentally sustainable products.

However, the positive relationship between product knowledge and attitudes toward purchasing recycled apparel was not supported. Many consumers remain uninformed regarding where to find sustainable fashion, often due to the scarcity of such products in the market [139,140]. This highlights the importance of product knowledge in building positive attitudes toward recycled clothing, suggesting that consumers' attitudes are more strongly influenced by their perceived value of these products. This could also imply that even if consumers are well informed about recycled clothing, this knowledge does not necessarily translate into a favorable attitude toward buying them. It is an interesting insight that suggests that other factors may play a more significant role in shaping attitudes toward the purchase of recycled apparel, such as cultural norms, social influences, perceived quality, or personal values. Understanding why product knowledge does not correlate with a positive attitude, in this case, could open up new avenues for research and action for marketers and sustainability advocates.

Next, consumers' perceptions of the benefits that they will obtain, including positive emotions (emotional value), an improved social life (social value), good quality (quality value), and good value for money (price value), all increase their positive attitudes toward purchasing recycled apparel. However, these results are in contrast to Sener et al.'s [13] study, which was conducted in Türkiye and found that social value and emotional value perceptions did not impact attitudes toward recycled fashion consumption. This discrep-

ancy highlights the cultural and market-specific differences in consumer attitudes toward recycled fashion, underscoring the need for targeted strategies in promoting sustainable fashion consumption.

Finally, the findings of this study also suggest that perceived emotional, social, quality, and price values all enhance the likelihood of purchase. These findings are consistent with past research [141–144] that underscores the positive impact of these value dimensions on sustainable purchasing behaviors. They suggest that the fashion industry should leverage these value dimensions in their design and marketing strategies.

Notably, the consumers surveyed in this study rated quality as the most crucial factor for recycled apparel. Consumer decision-making is typically intricate, with specific preferences impacting every facet of a product. In the world of recycled apparel, the value placed on meticulously crafted garments is significant. These findings reveal that high quality is essential, acting as a primary consideration for consumers contemplating the purchase of recycled clothing. This result also provides evidence that quality is the core content at the innermost level of a product concept.

Additionally, perceived price value had the second most important influence on purchase intention. However, Yoo et al. [145] reported that price value is one of the most important factors, as consumers are price-sensitive, even if they purchase sustainable products. Biswas and Roy [146] also found that price value was an important factor that had an impact on behavioral intention. Perceived social value also plays a key role, with the use of recyclable products potentially enhancing individuals' self-image as being environmentally conscious. This result underlines the needs of consumers for social approval and to make a good impression. Perceived emotional value, including comfort and satisfaction with one's self-image, further influences purchase intention. This result provides further support for the findings of a previous study, which implied that the consumption of recyclable and green products encourages consumers to consider themselves as being environmentally conscious by helping to protect the planet [147].

These results support the theory of perceived value, which emphasizes the role of perceived value in sustainable consumer behavior and suggests the need for tailored marketing strategies that highlight different value dimensions in different contexts.

6. Implications

6.1. Theoretical Implications

This study presents an integrated model combining the KAB and PCV frameworks to explain purchase intention toward recycled apparel. It enriches the literature in several key ways.

Firstly, by combining the PCV with the KAB model, this study illuminates the link between environmental concerns and the consumption of recycled fashion by considering both consumer-specific and product-specific factors. Environmental and product knowledge, as consumer-specific elements, alongside the PCV's product-specific factors, significantly enhance the model's capacity to bridge the attitude–behavior gap in sustainable fashion consumption.

Secondly, this research introduced and examined four distinct value dimensions social, price, emotional, and quality value—as metrics for perceived value. It uniquely identified the paramount importance of perceived quality value in recycled apparel, offering fresh insights into the persistent attitude–behavior gap and underlining the significance of tailoring perceived value considerations to specific product categories [36,148]. This aspect sheds light on the crucial value dimensions influencing consumer preferences for recycled apparel, providing valuable guidance for further research and practical applications in sustainable fashion [18,149].

Lastly, this study makes a pivotal contribution to the field of sustainable and recycled fashion, specifically within the non-Western context of China. It tackles the underexplored area of recycled apparel consumption in China, addressing a vital issue amid the global challenge of textile waste and the fashion industry's efforts to promote sustainable prac-

tices. By focusing on this specific geographical and cultural setting, the research fills a significant gap, offering a comprehensive understanding of consumer behavior toward recycled fashion in China and laying the groundwork for future studies and sustainable fashion initiatives.

6.2. Managerial Implications

This research sheds light on the decision-making process regarding recycled apparel, offering critical insights for fostering sustainable fashion consumption. It can support a multifaceted strategy to boost consumer interest in recycled apparel by engaging various stakeholders beyond merely fashion companies.

Enhancing consumers' environmental knowledge is essential. A joint effort between policymakers, educators, and media professionals to promote environmental awareness is crucial. Policymakers should introduce educational programs focusing on sustainability, while educators should incorporate these principles into their teaching. The media, especially social media platforms, should prioritize content that educates the public on environmental sustainability and the benefits of recycled and second-hand clothing, with the aim of increasing awareness and acceptance of these sustainable options.

Transparency and effective communication from retailers are key. Retailers should strive for openness about their sustainable practices, including the use of ecolabels and the promotion of their environmental initiatives. The media can support this by spotlighting these practices, thus bolstering consumers' trust in the fashion industry and its credibility.

Creating value through stakeholder collaboration is vital for aligning recycled apparel with consumer values. This requires a united effort from producers, marketers, and educators in communicating the multifaceted value of recycled apparel, enhancing the marketing strategies and educational programs.

Focusing on quality and fair pricing, companies should be transparent about the quality of their products and adopt pricing that reflects the true value of recycled apparel. Educators and policymakers can aid in this by educating consumers about the significance of quality and fair pricing in sustainable fashion.

Leveraging social and emotional values requires companies to collaborate with communities and influencers to create a sense of belonging and emotional connection to sustainable fashion. This can be achieved through community-focused content, influencer collaborations, and initiatives that engage consumers on an emotional level, including storytelling that underscores the environmental implications of fashion choices.

Promoting cross-sector partnerships for sustainable fashion is also critical. Encouraging collaborations among fashion companies, non-profits, educational institutions, and government bodies can lead to innovative solutions and a wider societal commitment to sustainable fashion. These partnerships should not only champion recycled apparel but also advocate for a broader spectrum of sustainable choices, including other eco-friendly products and practices.

By proposing these strategies, this study outlined a comprehensive approach to cultivating a more sustainable fashion industry, emphasizing the role of education, transparency, value creation, and collaboration in promoting recycled apparel and other sustainable fashion alternatives.

7. Limitations and Future Directions

This study, while insightful, presents several limitations. Firstly, it restricts the concept of perceived value (PV) to only four dimensions: social, emotional, functional, and price. Future studies could potentially explore and expand upon the dimensions of perceived value. Moreover, this study employs only two types of knowledge for analysis: environmental and product knowledge. Future research could benefit from the integration of other knowledge dimensions to enhance our understanding. It is important to consider that variables such as consumer demographics, their levels of involvement, and marketing strategies might also impact consumers' attitudes and behaviors. Future research could benefit for the several demographics is the several demographics.

efit from incorporating a broader range of dimensions and variables to capture the evolving trends in clothing consumption. Additionally, it would be valuable to explore the barriers and concerns that consumers face when choosing recycled clothing. Understanding these obstacles could provide deeper insights into consumer decision-making processes and help in developing more effective strategies to promote sustainable fashion consumption. We recognize that our study's focus on recycled apparel without encompassing all types of sustainable fashion products, such as upcycled clothing and second-hand garments, may limit the generalizability of our conclusions.

Another limitation of this study is its geographical scope. As it was conducted in China, its findings may not fully represent the perspectives of fashion consumers globally. Therefore, conducting similar research with a more diverse and larger sample group across different regions would enhance the generalizability of the results. Such an expanded study would not only validate the current findings but also offer a more comprehensive understanding of fashion consumers' attitudes and behaviors worldwide. Future research should consider incorporating a broader array of sustainable fashion products to offer a more comprehensive perspective and explore the role and effectiveness of different value dimensions in promoting sustainable fashion consumption.

Author Contributions: Writing—original draft, X.J. and K.F.; Writing—review & editing, A.O. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not Applicable.

Informed Consent Statement: Informed consent was obtained from all subjects involved in the study.

Data Availability Statement: The datasets used and/or analyzed during the current study are available from the corresponding author upon reasonable request.

Conflicts of Interest: The authors declare no conflicts of interest.

Appendix A. SmartPLS Graphical Presentations



Figure A1. PLS-SEM output for factor loadings, correlations, and coefficients of determination. EK for environmental knowledge, PK for product knowledge, PEV for perceived emotional value, PSV for perceived social value, PQV for perceived quality value, PPV for perceived price value, ATT for attitude, and PI for purchase intention.



Figure A2. PLS-SEM output for factor loadings and AVE. EK for environmental knowledge, PK for product knowledge, PEV for perceived emotional value, PSV for perceived social value, PQV for perceived quality value, PPV for perceived price value, ATT for attitude, and PI for purchase intention.



Figure A3. PLS-SEM output for factor loadings and CR. EK for environmental knowledge, PK for product knowledge, PEV for perceived emotional value, PSV for perceived social value, PQV for perceived quality value, PPV for perceived price value, ATT for attitude, and PI for purchase intention.



Figure A4. PLS-SEM output for path coefficients and t-values via bootstrapping procedure. EK for environmental knowledge, PK for product knowledge, PEV for perceived emotional value, PSV for perceived social value, PQV for perceived quality value, PPV for perceived price value, ATT for attitude, and PI for purchase intention.

References

- 1. Musova, Z.; Musa, H.; Drugdova, J.; Lazaroiu, G.; Alayasa, J. Consumer attitudes towards new circular models in the fashion industry. *J. Compet.* **2021**, *13*, 111. [CrossRef]
- Farhana, K.; Kadirgama, K.; Mahamude, A.S.F.; Mica, M.T. Energy consumption, environmental impact, and implementation of renewable energy resources in global textile industries: An overview towards circularity and sustainability. *Mater. Circ. Econ.* 2022, 4, 15. [CrossRef]
- 3. Stone, C.; Windsor, F.M.; Munday, M.; Durance, I. Natural or synthetic–how global trends in textile usage threaten freshwater environments. *Sci. Total Environ.* **2020**, *718*, 134689. [CrossRef]
- 4. Ikram, M. Transition toward green economy: Technological Innovation's role in the fashion industry. *Curr. Opin. Green Sustain. Chem.* **2022**, *37*, 100657. [CrossRef]
- 5. Centobelli, P.; Abbate, S.; Nadeem, S.P.; Garza-Reyes, J.A. Slowing the fast fashion industry: An all-round perspective. *Curr. Opin. Green Sustain. Chem.* 2022, *38*, 100684. [CrossRef]
- 6. China National Textile and Apparel Council. *Make Fashion Circular Outlook for a New Textile Economy in China;* China National Textile and Apparel Council: Beijing, China, 2021.
- 7. Ray, S.; Nayak, L. Marketing sustainable fashion: Trends and future directions. Sustainability 2023, 15, 6202. [CrossRef]
- 8. Wagner, M.M.; Heinzel, T. Human perceptions of recycled textiles and circular fashion: A systematic literature review. *Sustainability* **2020**, *12*, 10599. [CrossRef]
- 9. Chan, E.M.-H.; Cheung, J.; Leslie, C.A.; Lau, Y.-Y.; Suen, D.W.-S.; Tsang, C.-W. Revolutionizing the Textile and Clothing Industry: Pioneering Sustainability and Resilience in a Post-COVID Era. *Sustainability* **2024**, *16*, 2474. [CrossRef]
- 10. Jung, H.J.; Choi, Y.J.; Oh, K.W. Influencing factors of Chinese consumers' purchase intention to sustainable apparel products: Exploring consumer "attitude–behavioral intention" gap. *Sustainability* **2020**, *12*, 1770. [CrossRef]
- Su, J.; Watchravesringkan, K.; Zhou, J.; Gil, M. Sustainable clothing: Perspectives from US and Chinese young Millennials. *Int. J. Retail Distrib. Manag.* 2019, 47, 1141–1162. [CrossRef]
- 12. Botwinick, A.; Lu, S. Explore US retailers' merchandising strategies for clothing made from recycled textile materials. *Int. J. Fash. Des. Technol. Educ.* 2023, *16*, 131–140. [CrossRef]
- 13. Şener, T.; Bişkin, F.; Dündar, N. The effects of perceived value, environmental concern and attitude on recycled fashion consumption. *J. Fash. Mark. Manag. Int. J.* **2022**, *27*, 595–611. [CrossRef]
- 14. Kim, I.; Jung, H.J.; Lee, Y. Consumers' Value and Risk Perceptions of Circular Fashion: Comparison between Secondhand, Upcycled, and Recycled Clothing. *Sustainability* **2021**, *13*, 1208. [CrossRef]

- Lu, X.; Sheng, T.; Zhou, X.; Shen, C.; Fang, B. How Does Young Consumers' Greenwashing Perception Impact Their Green Purchase Intention in the Fast Fashion Industry? An Analysis from the Perspective of Perceived Risk Theory. *Sustainability* 2022, 14, 13473. [CrossRef]
- 16. Meng, M.D.; Leary, R.B. It might be ethical, but I won't buy it: Perceived contamination of, and disgust towards, clothing made from recycled plastic bottles. *Psychol. Mark.* 2021, *38*, 298–312. [CrossRef]
- 17. Pretner, G.; Darnall, N.; Testa, F.; Iraldo, F. Are consumers willing to pay for circular products? The role of recycled and second-hand attributes, messaging, and third-party certification. *Resour. Conserv. Recycl.* **2021**, 175, 105888. [CrossRef]
- 18. Chaturvedi, P.; Kulshreshtha, K.; Tripathi, V. Investigating the determinants of behavioral intentions of generation Z for recycled clothing: An evidence from a developing economy. *Young Consum.* **2020**, *21*, 403–417. [CrossRef]
- 19. Park, H.J.; Lin, L.M. Exploring attitude–behavior gap in sustainable consumption: Comparison of recycled and upcycled fashion products. *J. Bus. Res.* **2020**, *117*, 623–628. [CrossRef]
- 20. Kallgren, C.A.; Wood, W. Access to attitude-relevant information in memory as a determinant of attitude-behavior consistency. *J. Exp. Soc. Psychol.* **1986**, *22*, 328–338. [CrossRef]
- Predieri, S.; Cianciabella, M.; Daniele, G.M.; Gatti, E.; Lippi, N.; Magli, M.; Medoro, C.; Rossi, F.; Chieco, C. Italian Consumers' Awareness of Climate Change and Willingness to Pay for Climate-Smart Food Products. *Sustainability* 2023, 15, 4507. [CrossRef]
- Owojori, O.M.; Mulaudzi, R.; Edokpayi, J.N. Student's knowledge, attitude, and perception (KAP) to solid waste management: A survey towards a more circular economy from a rural-based tertiary institution in South Africa. *Sustainability* 2022, 14, 1310. [CrossRef]
- 23. Riskos, K.; Dekoulou, P.; Mylonas, N.; Tsourvakas, G. Ecolabels and the attitude–behavior relationship towards green product purchase: A multiple mediation model. *Sustainability* **2021**, *13*, 6867. [CrossRef]
- Hong, W.C.H.; Chi, C.; Liu, J.; Zhang, Y.; Lei, V.N.-L.; Xu, X. The influence of social education level on cybersecurity awareness and behaviour: A comparative study of university students and working graduates. *Educ. Inf. Technol.* 2023, 28, 439–470. [CrossRef]
- Zhao, Q.; Yu, X.; Feng, Y.; Zhao, Y.; Lu, J.; Guo, Y.; Li, X. Patients' and family members' knowledge, attitudes and perceived family support for SMBG in type 2 diabetes: A fuzzy-set qualitative comparative analysis. *BMJ Open* 2022, *12*, e063587. [CrossRef]
- 26. Sheth, J.N.; Newman, B.I.; Gross, B.L. Why we buy what we buy: A theory of consumption values. *J. Bus. Res.* **1991**, 22, 159–170. [CrossRef]
- Pereira, M.L.; de La Martinière Petroll, M.; Soares, J.C.; Matos, C.A.D.; Hernani-Merino, M. Impulse buying behaviour in omnichannel retail: An approach through the stimulus-organism-response theory. *Int. J. Retail Distrib. Manag.* 2023, 51, 39–58. [CrossRef]
- 28. Sebastián-Morillas, A.; Monfort, A.; López-Vázquez, B. Effects of perceived value and customer service on brand satisfaction. *J. Promot. Manag.* **2024**, *30*, 187–203. [CrossRef]
- 29. Boz, Z.; Korhonen, V.; Koelsch Sand, C. Consumer considerations for the implementation of sustainable packaging: A review. *Sustainability* **2020**, *12*, 2192. [CrossRef]
- Chen, H.-S.; Liang, C.-H.; Liao, S.-Y.; Kuo, H.-Y. Consumer attitudes and purchase intentions toward food delivery platform services. *Sustainability* 2020, 12, 10177. [CrossRef]
- 31. Wu, Y.; Huang, H. Influence of perceived value on consumers' continuous purchase intention in live-streaming e-commerce— Mediated by consumer trust. *Sustainability* **2023**, *15*, 4432. [CrossRef]
- 32. Sweeney, J.C.; Soutar, G.N. Consumer perceived value: The development of a multiple item scale. J. Retail. 2001, 77, 203–220. [CrossRef]
- 33. Polonsky, M.J.; Vocino, A.; Grau, S.L.; Garma, R.; Ferdous, A.S. The impact of general and carbon-related environmental knowledge on attitudes and behaviour of US consumers. *J. Mark. Manag.* **2012**, *28*, 238–263. [CrossRef]
- Taufique, K.M.R.; Vocino, A.; Polonsky, M.J. The influence of eco-label knowledge and trust on pro-environmental consumer behaviour in an emerging market. J. Strateg. Mark. 2017, 25, 511–529. [CrossRef]
- 35. Busalim, A.; Fox, G.; Lynn, T. Consumer behavior in sustainable fashion: A systematic literature review and future research agenda. *Int. J. Consum. Stud.* 2022, *46*, 1804–1828. [CrossRef]
- 36. Chi, T. Consumer perceived value of environmentally friendly apparel: An empirical study of Chinese consumers. *J. Text. Inst.* **2015**, *106*, 1038–1050. [CrossRef]
- 37. Diddi, S.; Yan, R.-N.; Bloodhart, B.; Bajtelsmit, V.; McShane, K. Exploring young adult consumers' sustainable clothing consumption intention-behavior gap: A Behavioral Reasoning Theory perspective. *Sustain. Prod. Consum.* **2019**, *18*, 200–209. [CrossRef]
- Okur, N.; Saricam, C.; Iri, A.; Sari, I. Analyzing the impact of COVID-19 on sustainable fashion consumption with a model based on consumer value perceptions. J. Fash. Mark. Manag. Int. J. 2023, 27, 826–850. [CrossRef]
- 39. Tanrikulu, C. Theory of consumption values in consumer behaviour research: A review and future research agenda. *Int. J. Consum. Stud.* **2021**, *45*, 1176–1197. [CrossRef]
- 40. Yoo, F.; Jung, H.J.; Oh, K.W. Motivators and barriers for buying intention of upcycled fashion products in China. *Sustainability* **2021**, *13*, 2584. [CrossRef]
- Sun, H.; Teh, P.L.; Linton, J.D. Impact of environmental knowledge and product quality on student attitude toward products with recycled/remanufactured content: Implications for environmental education and green manufacturing. *Bus. Strategy Environ.* 2018, 27, 935–945. [CrossRef]

- 42. Peña-García, N.; Gil-Saura, I.; Rodríguez-Orejuela, A.; Siqueira-Junior, J.R. Purchase intention and purchase behavior online: A cross-cultural approach. *Heliyon* 2020, *6*, e04284. [CrossRef]
- 43. Ajzen, I. The theory of planned behavior: Frequently asked questions. Hum. Behav. Emerg. Technol. 2020, 2, 314–324. [CrossRef]
- 44. Onofrei, G.; Filieri, R.; Kennedy, L. Social media interactions, purchase intention, and behavioural engagement: The mediating role of source and content factors. *J. Bus. Res.* **2022**, *142*, 100–112. [CrossRef]
- 45. Alwan, M.; Alshurideh, M. The effect of digital marketing on purchase intention: Moderating effect of brand equity. *Int. J. Data Netw. Sci.* **2022**, *6*, 837–848. [CrossRef]
- 46. Maria, S.; Pusriadi, T.; Hakim, Y.P.; Darma, D.C. The effect of social media marketing, word of mouth, and effectiveness of advertising on brand awareness and intention to buy. *J. Manaj. Indones.* **2019**, *19*, 107–122. [CrossRef]
- Ajzen, I.; Fishbein, M. Attitudes and the attitude-behavior relation: Reasoned and automatic processes. *Eur. Rev. Soc. Psychol.* 2000, 11, 1–33. [CrossRef]
- 48. Ajzen, I. The theory of planned behavior. Organ. Behav. Hum. Decis. Process. 1991, 50, 179-211. [CrossRef]
- 49. Al-Mamary, Y.H.S.; Alraja, M.M. Understanding entrepreneurship intention and behavior in the light of TPB model from the digital entrepreneurship perspective. *Int. J. Inf. Manag. Data Insights* **2022**, *2*, 100106. [CrossRef]
- 50. Wong, G.-Z.; Wong, K.-H.; Lau, T.-C.; Lee, J.-H.; Kok, Y.-H. Study of intention to use renewable energy technology in Malaysia using TAM and TPB. *Renew. Energy* 2024, 221, 119787. [CrossRef]
- 51. Mandarić, D.; Hunjet, A.; Vuković, D. The Impact of Fashion Brand Sustainability on Consumer Purchasing Decisions. *J. Risk Financ. Manag.* **2022**, *15*, 176. [CrossRef]
- 52. Wang, P.; Kuah, A.T.H.; Lu, Q.; Wong, C.; Thirumaran, K.; Adegbite, E.; Kendall, W. The impact of value perceptions on purchase intention of sustainable luxury brands in China and the UK. *J. Brand Manag.* **2021**, *28*, 325–346. [CrossRef]
- 53. Jain, S.; Mishra, S. Luxury fashion consumption in sharing economy: A study of Indian millennials. *J. Glob. Fash. Mark.* 2020, *11*, 171–189. [CrossRef]
- 54. Moon, K.; Lai, C.; Elita, L.; Chang, J. Popularization of sustainable fashion: Barriers and solutions. J. Text. Inst. 2014, 106, 939–952. [CrossRef]
- 55. Muposhi, A.; Chuchu, T. Influencing millennials to embrace sustainable fashion in an emerging market: A modified brand avoidance model perspective. *J. Fash. Mark. Manag. Int. J.* **2022**, 1–21. [CrossRef]
- 56. Fang, B. Environmental and Human Impacts of Fast Fashion. Commun. Humanit. Res. 2023, 15, 166–171. [CrossRef]
- 57. Breen, R. The Evil Fashion Giant: Shein and the Perpetuation of Overconsumption. Available online: https://opentext.ku.edu/environmentalgeopolitics/chapter/evil-fashion-giant-shein/ (accessed on 8 February 2024).
- 58. Abbate, S.; Centobelli, P.; Cerchione, R.; Nadeem, S.P.; Riccio, E. Sustainability trends and gaps in the textile, apparel and fashion industries. *Environ. Dev. Sustain.* 2024, 26, 2837–2864. [CrossRef]
- 59. Abrar, M.; Sibtain, M.M.; Shabbir, R. Understanding purchase intention towards eco-friendly clothing for generation Y & Z. *Cogent Bus. Manag.* 2021, *8*, 1997247. [CrossRef]
- 60. Hassan, S.H.; Yeap, J.A.L.; Al-Kumaim, N.H. Sustainable Fashion Consumption: Advocating Philanthropic and Economic Motives in Clothing Disposal Behaviour. *Sustainability* **2022**, *14*, 1875. [CrossRef]
- 61. Hossain, I.; Nekmahmud, M.; Fekete-Farkas, M. How Do Environmental Knowledge, Eco-Label Knowledge, and Green Trust Impact Consumers' Pro-Environmental Behaviour for Energy-Efficient Household Appliances? *Sustainability* **2022**, 14, 6513. [CrossRef]
- 62. Pratiwi, N.P.D.K.; Sulhaini, S.; Rinuastuti, B.H. The effect of environmental knowledge, green advertising and environmental attitude toward green purchase intention. *Russ. J. Agric. Socio-Econ. Sci.* **2018**, *78*, 95–105. [CrossRef]
- 63. Gkargkavouzi, A.; Halkos, G.; Matsiori, S. How do motives and knowledge relate to intention to perform environmental behavior? Assessing the mediating role of constraints. *Ecol. Econ.* **2019**, *165*, 106394. [CrossRef]
- 64. Amoah, A.; Addoah, T. Does environmental knowledge drive pro-environmental behaviour in developing countries? Evidence from households in Ghana. *Environ. Dev. Sustain.* **2021**, *23*, 2719–2738. [CrossRef]
- 65. Andika, A.; Bidayati, U. Unraveling the complexity of the organic food market: Indonesian consumer perspective on price and product knowledge. *Asian Manag. Bus. Rev.* **2024**, *4*, 73–89.
- 66. Lin, L.Y.; Chen, C.S. The influence of the country-of-origin image, product knowledge and product involvement on consumer purchase decisions: An empirical study of insurance and catering services in Taiwan. J. Consum. Mark. 2006, 23, 248–265. [CrossRef]
- 67. Wang, Y.; Hazen, B.T. Consumer product knowledge and intention to purchase remanufactured products. *Int. J. Prod. Econ.* **2016**, 181, 460–469. [CrossRef]
- 68. Alevizou, P.J.; Oates, C.J.; McDonald, S. The well(s) of knowledge: The decoding of sustainability claims in the UK and in Greece. *Sustainability* **2015**, *7*, 8729–8747. [CrossRef]
- 69. Samiee, S.; Chabowski, B.R. Knowledge structure in product-and brand origin–related research. *J. Acad. Mark. Sci.* 2021, 49, 947–968. [CrossRef]
- Kumar, A.; Prakash, G.; Kumar, G. Does environmentally responsible purchase intention matter for consumers? A predictive sustainable model developed through an empirical study. J. Retail. Consum. Serv. 2021, 58, 102270. [CrossRef]
- Dahlquist, S.H. How green product demands influence industrial buyer/seller relationships, knowledge, and marketing dynamic capabilities. J. Bus. Res. 2021, 136, 402–413. [CrossRef]

- 72. Nurhayati, T.; Hendar, H. Personal intrinsic religiosity and product knowledge on halal product purchase intention: Role of halal product awareness. *J. Islam. Mark.* **2020**, *11*, 603–620. [CrossRef]
- 73. Harahap, A.; Zuhriyah, A.; Rahmayanti, H.; Nadiroh, N. Relationship between knowledge of green product, social impact and perceived value with green purchase behavior. *E3S Web Conf.* **2018**, 74, 04002. [CrossRef]
- 74. Anggara, A.F.N.; Masnita, Y. Functional Theory of the Development of Religiosity and Knowledge on Intention to Purchase Bubble Tea Drinks in Indonesia. *J. Ilm. Manaj. Kesatuan* **2024**, *12*, 17–26. [CrossRef]
- 75. Sriminarti, N.; Nora, L. The role of product knowledge and attitudes toward purchasing intentions: The moderating effect of religiosity on halal cosmetics. *KnE Soc. Sci.* **2018**, *3*, 518–529. [CrossRef]
- 76. Zeithaml, V.A. Consumer perceptions of price, quality, and value: A means-end model and synthesis of evidence. *J. Mark.* **1988**, 52, 2–22. [CrossRef]
- Sánchez-Fernández, R.; Iniesta-Bonillo, M.Á. The concept of perceived value: A systematic review of the research. *Mark. Theory* 2007, 7, 427–451. [CrossRef]
- Liu, Y.; Yu, H.; Huang, H. The impact of purchase-decision involvement on purchasing intention: The mediating effect of customer perceived value. In Proceedings of the 2017 7th International Conference on Education and Management (ICEM 2017), Shenyang, China, 29–31 December 2017; pp. 850–854.
- 79. Raza, M.; Kolmakov, V.; Aziz, A.; Alshebami, A. The Impact of Consumer Perceived Value on Purchase Behaviour in a Developing Country: The Case of Green Products. *Creat. Innov. Manag.* **2020**, *13*, 80–99.
- 80. Yang, J.-J.; Ahn, S.-C. The effects of attitude, subjective norm, and behavioral intention on perceived values in traditional marketplaces. *J. Distrib. Sci.* 2020, *18*, 25–38.
- 81. Zhang, Y.; Xiao, C.; Zhou, G. Willingness to pay a price premium for energy-saving appliances: Role of perceived value and energy efficiency labeling. *J. Clean. Prod.* 2020, 242, 118555. [CrossRef]
- 82. Bloch, P.H. Seeking the ideal form: Product design and consumer response. J. Mark. 1995, 59, 16–29. [CrossRef]
- 83. Zheng, Q.; Zeng, H.; Xiu, X.; Chen, Q. Pull the emotional trigger or the rational string? A multi-group analysis of organic food consumption. *Foods* **2022**, *11*, 1375. [CrossRef]
- 84. Madadi, R.; Torres, I.M.; Zúñiga, M.Á. The semiotics of emojis in advertising: An integrated quantitative and qualitative examination of emotional versus functional ad dynamics. *Psychol. Mark.* **2024**. [CrossRef]
- 85. Lundblad, L.; Davies, I.A. The values and motivations behind sustainable fashion consumption. *J. Consum. Behav.* **2016**, *15*, 149–162. [CrossRef]
- 86. Jia, T.; Iqbal, S.; Ayub, A.; Fatima, T.; Rasool, Z. Promoting responsible sustainable consumer behavior through sustainability marketing: The boundary effects of corporate social responsibility and brand image. *Sustainability* **2023**, *15*, 6092. [CrossRef]
- 87. Gonçalves, H.M.; Lourenço, T.F.; Silva, G.M. Green buying behavior and the theory of consumption values: A fuzzy-set approach. *J. Bus. Res.* **2016**, *69*, 1484–1491. [CrossRef]
- 88. Özdil, M. The effect of clothing as a marker on identity. Motif Akad. Halk Bilim. Derg. 2021, 14, 117–130. [CrossRef]
- 89. Crane, D. Fashion and Its Social Agendas: Class, Gender, and Identity in Clothing; University of Chicago Press: Chicago, IL, USA, 2000.
- 90. Entwistle, J. The Fashioned Body: Fashion, Dress and Modern Social Theory; John Wiley & Sons: Hoboken, NJ, USA, 2023.
- 91. Barnard, M. Fashion as Communication; Routledge: London, UK, 2013.
- Gwozdz, W.; Gupta, S.; Gentry, J. The Role of Fashion vs. Style Orientation on Sustainable Apparel Consumption: An Update. In Proceedings of the 41st Annual Macromarketing Conference: Macromarketing and Academic Activism, Dublin, Ireland, 13–15 July 2016; pp. 639–660.
- 93. Sarokin, S.N.; Bocken, N.M.P. Pursuing profitability in slow fashion: Exploring brands' profit contributors. J. Clean. Prod. 2024, 444, 141237. [CrossRef]
- 94. Vu, N.T.; Anh, M.H. Consumer Perception and Behavior in the Context of Sustainable Fashion: A Comprehensive Analysis of Emerging Trends and Impacts. *J. Hum. Behav. Soc. Sci.* **2019**, *3*, 1–14.
- 95. Wasaya, A.; Saleem, M.A.; Ahmad, J.; Nazam, M.; Khan, M.M.A.; Ishfaq, M. Impact of green trust and green perceived quality on green purchase intentions: A moderation study. *Environ. Dev. Sustain.* **2021**, *23*, 13418–13435. [CrossRef]
- 96. Queiroz, F.C.B.P.; Lima, N.C.; da Silva, C.L.; Queiroz, J.V.; de Souza, G.H.S. Purchase intentions for brazilian recycled PET products—Circular economy opportunities. *Recycling* **2021**, *6*, 75. [CrossRef]
- 97. Cheung, R.; Lam, A.Y.C.; Lau, M.M. Drivers of green product adoption: The role of green perceived value, green trust and perceived quality. *J. Glob. Sch. Mark. Sci.* 2015, 25, 232–245. [CrossRef]
- 98. Tang, H.; Rasool, Z.; Khan, M.A.; Khan, A.I. The Effect of Green Percieved Quality & Risk, Green Trust on Consumer's and Green Purchased Intention with a Moderation of Environmental Awareness; Research Square: Durham, NC, USA, 2022.
- 99. Hasbullah, N.N.; Sulaiman, Z.; Mas'od, A. The effect of perceived value on sustainable fashion consumption in the era of COVID-19: A proposed conceptual framework. *Int. J. Acad. Res. Bus. Soc. Sci* **2020**, *1*, 895–906. [CrossRef] [PubMed]
- 100. Khan, S.; Zaman, S.I.; Rais, M. Measuring student satisfaction through overall quality at business schools: A structural equation modeling: Student satisfaction and quality of education. *S. Asian J. Soc. Rev.* **2022**, *1*, 34–55. [CrossRef]
- Jeaheng, Y.; Al-Ansi, A.; Han, H. Impacts of Halal-friendly services, facilities, and food and Beverages on Muslim travelers' perceptions of service quality attributes, perceived price, satisfaction, trust, and loyalty. J. Hosp. Mark. Manag. 2020, 29, 787–811. [CrossRef]

- 102. Singh, S.; Alok, S. Drivers of repurchase intention of organic food in India: Role of perceived consumer social responsibility, price, value, and quality. *J. Int. Food Agribus. Mark.* 2022, 34, 246–268. [CrossRef]
- 103. Al Mamun, A.; Rahman, M.K.; Masud, M.M.; Mohiuddin, M. Willingness to pay premium prices for green buildings: Evidence from an emerging economy. *Environ. Sci. Pollut. Res.* **2023**, *30*, 78718–78734. [CrossRef] [PubMed]
- Carrión Bósquez, N.G.; Arias-Bolzmann, L.G.; Martínez Quiroz, A.K. The influence of price and availability on university millennials' organic food product purchase intention. *Br. Food J.* 2023, 125, 536–550. [CrossRef]
- Malhotra, G.; Srivastava, H. Green purchase intention in omnichannel retailing: Role of environmental knowledge and price sensitivity. J. Strateg. Mark. 2023, 1–24. [CrossRef]
- 106. Ito, K.; Zhang, S. Willingness to pay for clean air: Evidence from air purifier markets in China. *J. Political Econ.* **2020**, *128*, 1627–1672. [CrossRef]
- 107. Shen, B.; Cao, Y.; Xu, X. Product line design and quality differentiation for green and non-green products in a supply chain. *Int. J. Prod. Res.* **2020**, *58*, 148–164. [CrossRef]
- Bigliardi, B.; Campisi, D.; Ferraro, G.; Filippelli, S.; Galati, F.; Petroni, A. The intention to purchase recycled products: Towards an integrative theoretical framework. *Sustainability* 2020, *12*, 9739. [CrossRef]
- Likert, R. A Technique for the Measurement of Attitudes. Arch. Psychol. 1932. Available online: https://legacy.voteview.com/ pdf/Likert_1932.pdf (accessed on 25 January 2024).
- 110. Brislin, R.W. Back-translation for cross-cultural research. J. Cross-Cult. Psychol. 1970, 1, 185–216. [CrossRef]
- 111. Kim, H.-S.; Damhorst, M.L. Environmental concern and apparel consumption. *Cloth. Text. Res. J.* 1998, 16, 126–133. [CrossRef]
- 112. Kang, J.; Liu, C.; Kim, S.H. Environmentally sustainable textile and apparel consumption: The role of consumer knowledge, perceived consumer effectiveness and perceived personal relevance. *Int. J. Consum. Stud.* **2013**, *37*, 442–452. [CrossRef]
- 113. Walsh, G.; Shiu, E.; Hassan, L.M. Replicating, validating, and reducing the length of the consumer perceived value scale. *J. Bus. Res.* 2014, 67, 260–267. [CrossRef]
- Preuit, R.; Yan, R.-N.T. Fashion and sustainability: Increasing knowledge about slow fashion through an educational module. In Proceedings of the International Textile and Apparel Association Annual Conference Proceedings, Vancouver, BC, Canada, 8–11 November 2016.
- 115. Bell, E.; Bryman, A.; Harley, B. Business Research Methods; Oxford University Press: Oxford, UK, 2022.
- 116. Wong, K.K.-K. Mastering Partial Least Squares Structural Equation Modeling (PLS-Sem) with Smartpls in 38 Hours; IUniverse: Bloomington, India, 2019.
- 117. Sarstedt, M.; Ringle, C.M.; Smith, D.; Reams, R.; Hair, J.F., Jr. Partial least squares structural equation modeling (PLS-SEM): A useful tool for family business researchers. J. Fam. Bus. Strategy 2014, 5, 105–115. [CrossRef]
- 118. Hair, J.F., Jr.; Hair, J.; Sarstedt, M.; Ringle, C.M.; Gudergan, S.P. Advanced Issues in Partial Least Squares Structural Equation Modeling; Sage Publications: Thousand Oaks, CA, USA, 2023.
- Hair, J.F., Jr.; Matthews, L.M.; Matthews, R.L.; Sarstedt, M. PLS-SEM or CB-SEM: Updated guidelines on which method to use. Int. J. Multivar. Data Anal. 2017, 1, 107–123. [CrossRef]
- 120. Hair, J.F., Jr.; Sarstedt, M.; Hopkins, L.; Kuppelwieser, V.G. Partial least squares structural equation modeling (PLS-SEM) An emerging tool in business research. *Eur. Bus. Rev.* 2014, *26*, 106–121. [CrossRef]
- 121. Yang, J.; Al Mamun, A.; Reza, M.N.H.; Yang, M.; Abd Aziz, N. Predicting the significance of consumer environmental values, beliefs, and norms for sustainable fashion behaviors: The case of second-hand clothing. *Asia Pac. Manag. Rev.* 2024; *in press.* [CrossRef]
- 122. Usman, H.; Projo, N.W.K.; Chairy, C.; Haque, M.G. The role of trust and perceived risk on Muslim behavior in buying halalcertified food. *J. Islam. Mark.* 2024; *ahead-of-print.* [CrossRef]
- 123. Yang, D.; Fang, N.; Ramirez-Asis, E.; Alashker, Y.; Abourehab, M.A.S.; Zhang, K. Mediating Role of Risk Perception and Environmental Quality on the Relationship Between Risk Knowledge and Traveler's Intention in COVID-19. *Front. Environ. Sci.* 2022, 10, 902457. [CrossRef]
- 124. Hair, J.F., Jr.; Hult, G.T.M.; Ringle, C.M.; Sarstedt, M.; Danks, N.P.; Ray, S. Partial Least Squares Structural Equation Modeling (PLS-SEM) Using R: A Workbook; Springer: Berlin/Heidelberg, Germany, 2021.
- 125. Memon, M.A.; Ramayah, T.; Cheah, J.-H.; Ting, H.; Chuah, F.; Cham, T.H. PLS-SEM statistical programs: A review. *J. Appl. Struct. Equ. Model.* **2021**, *5*, 1–14. [CrossRef]
- 126. Hair, J.F.; Risher, J.J.; Sarstedt, M.; Ringle, C.M. When to use and how to report the results of PLS-SEM. *Eur. Bus. Rev.* 2019, *31*, 2–24. [CrossRef]
- Henseler, J.; Ringle, C.M.; Sarstedt, M. A new criterion for assessing discriminant validity in variance-based structural equation modeling. J. Acad. Mark. Sci. 2015, 43, 115–135. [CrossRef]
- Franke, G.; Sarstedt, M. Heuristics versus statistics in discriminant validity testing: A comparison of four procedures. *Internet Res.* 2019, 29, 430–447. [CrossRef]
- 129. Falk, R.F.; Miller, N.B. A Primer for Soft Modeling; University of Akron Press: Akron, OH, USA, 1992.
- 130. Geisser, S. A predictive approach to the random effect model. Biometrika 1974, 61, 101–107. [CrossRef]
- Purwanto, A. Partial Least Squares Structural Squation Modeling (PLS-SEM) Analysis for Social and Management Research: A Literature Review. J. Ind. Eng. Manag. Res. 2021. Available online: https://ssrn.com/abstract=3982764 (accessed on 1 February 2024).

- 132. Witte, R.S.; Witte, J.S. Statistics; John Wiley & Sons: Hoboken, NJ, USA, 2017.
- 133. Johnson, R.A.; Wichern, D.W. Applied Multivariate Statistical Analysis; Pearson Education Limited: London, UK, 2002.
- 134. Latan, H.; Noonan, R.; Matthews, L. Partial least squares path modeling. In *Partial Least Squares Path Modeling: Basic Concepts, Methodological Issues and Applications*; Springer: Berlin/Heidelberg, Germany, 2017.
- 135. Jalil, M.H.; Shaharuddin, S.S. Consumer purchase behavior of eco-fashion clothes as a trend to reduce clothing waste. *Int. J. Innov. Technol. Explor. Eng.* **2019**, *8*, 4224–4233. [CrossRef]
- 136. Bong Ko, S.; Jin, B. Predictors of purchase intention toward green apparel products: A cross-cultural investigation in the USA and China. *J. Fash. Mark. Manag. Int. J.* **2017**, *21*, 70–87. [CrossRef]
- Vicente-Molina, M.A.; Fernández-Sáinz, A.; Izagirre-Olaizola, J. Environmental knowledge and other variables affecting proenvironmental behaviour: Comparison of university students from emerging and advanced countries. *J. Clean. Prod.* 2013, *61*, 130–138. [CrossRef]
- 138. Jaiswal, D.; Kant, R. Green purchasing behaviour: A conceptual framework and empirical investigation of Indian consumers. *J. Retail. Consum. Serv.* **2018**, *41*, 60–69. [CrossRef]
- 139. Gleim, M.; Lawson, S.J. Spanning the gap: An examination of the factors leading to the green gap. *J. Consum. Mark.* 2014, *31*, 503–514. [CrossRef]
- 140. Saepudin, D.; Shojaei, A.S.; Barbosa, B.; Pedrosa, I. Intention to Purchase Eco-Friendly Handcrafted Fashion Products for Gifting and Personal Use: A Comparison of National and Foreign Consumers. *Behav. Sci.* 2023, 13, 171. [CrossRef]
- 141. Li, Z.; Shu, S.; Shao, J.; Booth, E.; Morrison, A.M. Innovative or not? The effects of consumer perceived value on purchase intentions for the palace museum's cultural and creative products. *Sustainability* **2021**, *13*, 2412. [CrossRef]
- 142. Tahir, M. The impact of green perceived value and its additional multidimensional expanded variables effect on customer attitude and purchasing intention for buying green food products: A case of Pakistan. J. Organ. Bus. 2021, 2, 24–42.
- Alsaid, K.N.; Saleh, M.A.H. Perceived value and purchase intention of counterfeit luxury brands: Testing the moderation of materialism. *Amity J. Mark.* 2019, 4, 1–17.
- 144. Hamari, J.; Hanner, N.; Koivisto, J. "Why pay premium in freemium services?" A study on perceived value, continued use and purchase intentions in free-to-play games. *Int. J. Inf. Manag.* **2020**, *51*, 102040. [CrossRef]
- 145. Yoo, J.-J.; Divita, L.; Kim, H.-Y. Environmental awareness on bamboo product purchase intentions: Do consumption values impact green consumption? *Int. J. Fash. Des. Technol. Educ.* 2013, *6*, 27–34. [CrossRef]
- 146. Biswas, A.; Roy, M. Leveraging factors for sustained green consumption behavior based on consumption value perceptions: Testing the structural model. *J. Clean. Prod.* **2015**, *95*, 332–340. [CrossRef]
- 147. Bei, L.-T.; Simpson, E.M. The Determinants of Consumers' Purchase Decisions for Recycled Products: An Application of Acquisition-Transaction Utility Theory. *Adv. Consum. Res.* **1995**, *22*, 257.
- 148. Lou, X.; Chi, T.; Janke, J.; Desch, G. How Do Perceived Value and Risk Affect Purchase Intention toward Second-Hand Luxury Goods? An Empirical Study of U.S. Consumers. *Sustainability* **2022**, *14*, 11730. [CrossRef]
- 149. Yu, S.; Lee, J. The effects of consumers' perceived values on intention to purchase upcycled products. *Sustainability* **2019**, *11*, 1034. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.